MATH333: Tutorial 9 (7-4-00)

1 Correlation Dimension Of Fractals

Please try to read the programmes given to you. I will discuss those programmes in the coming tutorial. The programmes will be very useful for your first project after you obtained the Poincare Section, if any, of the system you choose to study.

2 IMPORTANT NOTE

Remind you all again that you HAVE to start to do your project NOW. No matter in the form of reference book searching, or program writing. You will NOT be able to finish it if you still haven’t started it yet. Remember that there is still only less than 2 weeks left before the due date. No late paper will be accepted. Please don’t under-estimate the work load of this paper.